Comments from reviewers:

Reviewer #1:
What was the parenchymal thickness of the right kidney? Was there any renal impairment on the right side?

Response: Thank you for pointing this out. In the previous manuscript, we referred that “There was no change in creatinine clearance in the serial examinations”, the result of urine test and renal ultrasound (parenchymal thickness 2.0 cm) have been added in the revised manuscript.

Right side hydronephrosis during pregnancy might be related to the hormonal effects of pregnancy and mechanical compression from the growing uterus. Operative technique is a bit detailed and I think needs to be summarized. What was the need to do augmentation so expeditiously? Why not decompress the system with nephrostomy and operate until after delivery?

Response: As was mentioned in “Timeline of the case” of our article, The patient underwent augmentation sigmoidocolo cystoplasty at 21y/o (2010, 9 years before pregnancy), and she was pregnant at 30y/o (2019). Our team quite agree with reviewer at the point that decompressing the system with nephrostomy is safe before delivery and can protect the isolated kidney when bladder deformity and the trabecula made it difficult to find the ureteric orifice to place a double pigtail ureteral stent under cystoscopy. Actually, we performed percutaneous nephrostomy drainage because of acute pyelonephritis in the right kidney and progressive hydronephrosis in the second trimester.

Reviewer #2:
Some new literatures were not reviewed. Authors should update the reference.

Response: Corrected as suggested, thank you. We have updated our references as much as possible. More keywords were used to search, and all the literatures found in each relevant article have also been included, especially the new reference. 8 other articles have been added in the revised manuscript.
Comments from re-reviewer:
Authors reported a case of pregnancy and delivery with success after augmentation cystoplasty. Revised manuscript was well-written. But several spelling errors or grammatical errors were found. It is acceptable after English check.
Response: Thank you for your comment. We have revised spelling errors or grammatical errors and check the English again.